\documentclass[11pt]{article}
\usepackage[latin1]{inputenc}
\usepackage{amsmath,amsthm,amssymb}
\usepackage{amsfonts}
\usepackage{amsmath,amsthm,amssymb,amscd}
\usepackage{latexsym}
\usepackage{graphicx}
\usepackage{mathrsfs}
\usepackage{graphicx, color}
\usepackage{amscd}
\usepackage{amsmath}
\usepackage{amsfonts}
\usepackage{amssymb}
\usepackage{mathrsfs}
%\usepackage[notref,notcite]{showkeys}
 \font\rmg=cmr10
\font\bfg=cmbx10
\renewcommand{\arraystretch}{1.5}
\textwidth172mm \textheight22cm \hoffset-24mm \voffset-20mm
\font\sans=cmss10
%\textwidth=13.5cm
%\baselineskip=16pt
\makeatletter \@addtoreset{equation}{section} \makeatother
\renewcommand{\theequation}{\arabic{section}.\arabic{equation}}
\setlength{\parindent}{1em}
\newcommand{\zwfs}[2]{\frac{\;#1\;}{\;#2\;}}
\newtheorem{theorem}{Theorem}[section]
\newtheorem{definition}{Definition}[section]
\newtheorem{proposition}{Proposition}[section]
\newtheorem{lemma}{Lemma}[section]
\newtheorem{remark} {Remark}[section]
\newtheorem{example}{Example}[section]
%\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{corollary}[theorem]{Corollary}
%\newtheorem{proposition}[theorem]{Proposition}
\def\qed{\hfill $\Box$\par\vskip3mm}

\begin{document}
\begin{thebibliography}{99}

\bibitem{r22} R. A. Adams, J. J. F. Fournier, {\it Sobolev Spaces}, Academic Press, New York-London, 2003.

\bibitem{r36} S. Alama, G. Tarantello, Elliptic problems with nonlinearities indefinite in sign, {\it J. Funct. Anal.} {\bf 141} (1996), 159--215.

\bibitem{ABC} A. Ambrosetti, H. Brezis, G. Cerami, Combined effects of concave-convex nonlinearities in some elliptic problems, {\it J. Funct. Anal.} {\bf 122} (1994), 519--543.

\bibitem{r2} D. Applebaum, L\'{e}vy processes--from probability to finance quantum groups,
{\it Notices Amer. Math. Soc.} {\bf 51} (2004), 1336--1347.

\bibitem{r35} G. Autuori, P. Pucci, Elliptic problems involving the fractional Laplacian in $\mathbb{R}^N$, {\it J. Differential Equations} {\bf 255} (2013), 2340--2362.

\bibitem{r24} G. Autuori, P. Pucci, Existence of entire solutions for a class of quasilinear elliptic equations, {\it  NoDEA Nonlinear Differential Equations Appl.} {\bf 20} (2013), 977--1009.

\bibitem{r4} B. Barrios, E. Colorado, A. De Pablo, U. Sanchez, On some critical problems for the fractional Laplacian operator, {\it J. Differential Equations} {\bf 252} (2012), 6133--6162.

\bibitem{r21} H. Brezis, E. Lieb, A relation between pointwise convergence of functionals and convergence of functionals, {\it Proc. Amer. Math. Soc.} {\bf 88} (1983), 486--490.


\bibitem{r-1} L. Caffarelli, Nonlocal diffusions, drifts and games, {\it Nonlinear Partial Differential Equations, Abel Symposia} {\bf 7} (2012), 37--52.

\bibitem{r1} L. Caffarelli, L. Silvestre, An extension problem related to the fractional Laplacian, {\it Comm. Partial Differential Equations} {\bf 32} (2007), 1245--1260.

\bibitem{r33} X. Chang, Z. Q. Wang, Ground state of scalar field equations involving a fractional Laplacian with general nonlinearity, {\it Nonlinearity} {\bf 26} (2013), 479--494.

\bibitem{CW} X. Chang, Z. Q. Wang, Nodal and multiple solutions of nonlinear problems involving the fractional Laplacian, {\it J. Differential Equations} {\bf
256} (2014), 2965--2992.

\bibitem{r} M. Cheng, Bound state for the fractional Schr\"{o}dinger equation with unbounded potential, {\it J. Math. Phys.} {\bf 53} (2012) 043507, 7pp.

\bibitem{r32} S. Dipierro, G. Palatucci, E. Valdinoci, Existence and symmetry results for a  Schr\"{o}dinger type problem involving the fractional Laplacian, {\it Matematiche} 68 (2013), 201--216.

\bibitem{r8} E. Di Nezza, G. Palatucci, E. Valdinoci, Hitchhiker's guide to the fractional Sobolev spaces, {\it Bull. Sci. Math.} {\bf 136} (2012), 521--573.

\bibitem{r31} P. Felmer, A. Quaas, J. Tan, Positive solutions of the nonlinear Schr\"{o}dinger equation with the fractional Laplacian, {\it Proc. Roy. Soc. Edinburgh Sect. A} {\bf 142} (2012), 1237--1262.


\bibitem{r-8} M. Ferrara, G. Molica Bisci, B.L. Zhang, Existence of weak solutions for non-local fractional problems via Morse theory, {\it Discrete Contin. Dyn. Syst. Ser. B} {\bf 19} (2014), 2483--2499.

\bibitem{r10} A. Iannizzotto, S. Liu, K. Perera, M. Squassina, Existence results for fractional $p$-Laplacian problems via Morse theory, {\it Advances in Calculus of Variations}, doi:10.1515/acv-2014-0024.


\bibitem{r12} A. Iannizzotto, M. Squassina, Weyl--type laws for fractional $p$-eigenvalue problems, {\it Asymptotic Anal.} {\bf 88} (2014), 233--245.

\bibitem{L1} N. Laskin, Fractional quantum mechanics and  L\'evy path integrals, {\it Phys. Lett. A} {\bf 268} (2000), 298--305.

\bibitem{L2} N. Laskin, Fractional Schr\"odinger equation, {\it Phys. Rev. E} {\bf 66} (2002) 056108, 18 pp.

\bibitem{r11}  E. Lindgren, P. Lindqvist, Fractional eigenvalues, {\it Calc. Var.  Partial Differential Equations} {\bf 49} (2014), 795--826.

\bibitem{r-3} R. Metzler, J. Klafter, The restaurant at the random walk: recent developments in the description of anomalous transport by fractional dynamics, {\it J. Phys. A} {\bf 37} (2004), 161--208.


\bibitem{MRS}  G. Molica Bisci, V. R\u adulescu, R. Servadei, {\it Variational Methods for Nonlocal
Fractional Problems}, Cambridge University Press, Cambridge, 2016.

\bibitem{r-4} G. Molica Bisci, Fractional equations with bounded primitive, {\it Appl. Math. Lett.} {\bf 27} (2014), 53--58.

\bibitem{M} G. Molica Bisci, V. R\u adulescu, Ground state solutions of scalar field fractional for Schr\"{o}dinger equations,
{\it Calc. Var. Partial Differential Equations} (2015), doi :10.1007/s00526-015-0891-5.

\bibitem{MR} G. Molica Bisci, D. Repovs, Higher nonlocal problems with bounded potential, {\it J. Math. Anal. Appl.} {\bf 420} (2014), 591--601.

\bibitem{PR} P. Pucci, V. R\u{a}dulescu, Combined effects in quasilinear elliptic problmes with lack of compactness, {\it Rend. Lincei Mat. Appl.} {\bf 22} (2011), 189--205.

\bibitem{PZ} P. Pucci, Q. Zhang, Existence of entire solutions for a class of variable exponent elliptic equations, {\it J. Differential Equations} {\bf
257} (2014), 1529--1566.

\bibitem{PXZ1} P. Pucci, M.Q. Xiang, B.L. Zhang, Multiple solutions for nonhomogeneous Schr\"{o}dinger--Kirchhoff
type equations involving the fractional $p$--Laplacian in $\mathbb{R}^N$, {\it Calc. Var. Partial Differential Equations} (2015),
doi: 10.1007/s00526-015-0883-5.

\bibitem{PXZ2} P. Pucci, M.Q. Xiang, B.L. Zhang, Existence and multiplicity of entire solutions for fractional $p$-Kirchhoff equations,
{\it Adv. Nonlinear Anal.}, to appear.

\bibitem{r34} S. Secchi, Ground state solutions for the fractional Schr\"{o}dinger in $\mathbb{R}^N$, {\it J. Math. Phys.} {\bf 54} (2013) 031501, 17pp.

\bibitem{r3} R. Servadei, E. Valdinoci, The Brezis--Nirenberg result for the fractional Laplacian, {\it Trans. Amer. Math. Soc.} {\bf 367} (2015),  67--102.

\bibitem{r17} R. Servadei, E. Valdinoci, On the spectrum of two different fractional operators, {\it Proc. Roy. Soc. Edinburgh Sect.
A} {\bf 144} (2014), 831--855.

\bibitem{r14} R. Servadei, E. Valdinoci, Mountain pass solutions for non-local elliptic operators, {\it J. Math. Anal. Appl.} {\bf 389} (2012), 887--898.

\bibitem{r23}  M. Struwe, {\it Variational Methods: Applications to Nonlinear Partial Differential Equations and Hamiltonian Systems}, Springer-Verlag, Berlin-Heidelberg, 1990.

\bibitem{XZF} M.Q. Xiang, B.L. Zhang, M. Ferrara, Multiplicity results for the nonhomogeneous
fractional $p$-Kirchhoff equations with concave--convex nonlinearities, {\it Proc. Roy. Soc. A} (2015), doi: 10.1098/rspa.2015.0034.

\bibitem{XZG}
M.Q. Xiang, B.L. Zhang, X.Y. Guo, Infinitely many solutions for a fractional Kirchhoff type problem via fountain theorem,
{\it Nonlinear Anal.} 120 (2015), 299--313.


\end{thebibliography}

\title{Existence of solutions for perturbed fractional $p$-Laplacian equations}
\author{ Mingqi Xiang$^a$,\  Binlin Zhang$^{b, }$\footnote{Corresponding author.
E-mail address:\, {\tt xiangmingqi\_hit@163.com} (M. Xiang),\, {\tt zhangbinlin2012@163.com} (B. Zhang), {\tt vicentiu.radulescu@imar.ro} (V. R\u adulescu) } \ and\ Vicen\c tiu D. R\u adulescu$^{c,d}$\\
\footnotesize $^a$ Department of Mathematics, Civil Aviation University of China, Tianjin,
300300, P.R. China\\
\footnotesize $^b$ Department of Mathematics, Heilongjiang Institute of Technology, Harbin,
150050, P.R. China\\
\footnotesize $^c$ Department of Mathematics, Faculty of Sciences, King Abdulaziz University,\\ \footnotesize
P.O. Box 80203, Jeddah 21589, Saudi Arabia\\
\footnotesize $^d$ Institute of Mathematics ``Simion Stoilow" of the Romanian Academy,\\ \footnotesize P.O. Box 1-764,
014700 Bucharest, Romania
}
\date{ }
\maketitle


\begin{abstract}
{ The purpose of this paper is to investigate the existence of weak solutions for a perturbed nonlinear elliptic equation driven by the fractional $p$-Laplace operator as follows:
\begin{align*}
(-\Delta)^s_pu+V(x)|u|^{p-2}u=\lambda a(x)|u|^{r-2}u-b(x)|u|^{q-2}u\quad &\mbox{in }\,\,\mathbb{R}^N,
\end{align*}
where $\lambda$ is a real parameter, $(-\Delta )^s_p$ is the fractional $p$-Laplacian operator with $0<s<1<p<\infty$, $p<r<\min\{q,p_s^*\}$ and $V, a, b:\mathbb{R}^N\rightarrow(0,\infty)$ are three positive weights. Using variational methods, we obtain nonexistence and multiplicity results
for the above-mentioned equations depending on $\lambda$ and according to the integrability properties of the ratio $a^{q-p}/b^{r-p}$. Our results
extend the previous work of Autuori and Pucci (2013) \cite{r35}} to the fractional $p$-Laplacian setting. Furthermore, we weaken one of the conditions used in their paper. Hence the results of this paper are new even in the fractional Laplacian case.
\\
\emph{\bf Keywords:}  fractional $p$-Laplacian; fractional Schr\"{o}dinger equation; mountain pass theorem; variational methods.\\
\emph{\bf 2010 MSC:} 35R11, 35A15, 35J60, 47G20.
\end{abstract}

%{\section*{{\bf 1. Introduction }}} \setcounter{section}{1}
 %\setcounter{equation}{0}
\section{Introduction}

In this paper we deal with the following one parameter elliptic equations:
\begin{eqnarray}\label{k1}
(-\Delta)^s_pu+V(x)|u|^{p-2}u=\lambda a(x)|u|^{r-2}u-b(x)|u|^{q-2}u & \mbox{in } \mathbb{R}^N,
\end{eqnarray}
where $N> ps$ with $s\in(0,1)$ and $(-\Delta)_p^{s}$ is the fractional $p-$Laplace operator
which, up to normalization factors, may be defined as
\begin{equation*}
(-\Delta)^s_p u(x)=2\lim_{\varepsilon\searrow0}
\int_{\mathbb{R}^{N}\backslash B_\varepsilon(x)}\frac{|u(x)-u(y)|^{p-2}(u(x)-u(y))}{|x-y|^{N+ps}}\,dy
\end{equation*}
for $x\in\mathbb{R}^{N}$, where $B_\varepsilon(x):=\{y\in \mathbb{R}^N: |x-y|<\varepsilon\}$. As for some recent results on the fractional $p$-Laplacian, we refer to for example \cite{r10, r12, r11} and the references therein.




Notice that when $p=2$, problem \eqref{k1} reduces to the following fractional Laplacian equations:
\begin{eqnarray}
(-\Delta)^{s} u+V(x)u=\lambda a(x)|u|^{r-2}u-b(x)|u|^{q-2}u  & \mbox{in } \mathbb{R}^N,
\end{eqnarray}
which can been seen as the fractional form of the following classical stationary Schr\"{o}dinger equations:
\begin{eqnarray}
-\Delta u+V(x)u= \lambda a(x)|u|^{r-2}u-b(x)|u|^{q-2}u & \mbox{in } \mathbb{R}^N.
\end{eqnarray}
For standing wave solutions of fractional Schr\"{o}dinger equations in $\mathbb{R}^N$, we refer to \cite{r33, r32, r31, L1, L2} and the references therein. Especially, models governed by unbounded potentials involving fractional Schr\"{o}dinger equations have been investigated in the last years, see for instance \cite{r, r34}.

However, all these papers deal with problems which are not directly comparable to problem \eqref{k1}. In fact,
the present paper is inspired by the the following works:
{\em Alama} and {\em Tarantello} in \cite{r36} studied the following Dirichlet problem with indefinite weights:
\begin{eqnarray*}
-\Delta u-\lambda u=\omega(x)u^{q-1}-h(x)u^{r-1}\ \ &{\rm in}\ \Omega,
\end{eqnarray*}
where $\lambda\in\mathbb{R},\ \Omega\subset\mathbb{R}^N$ ($N\geq3$) is a bounded domain with smooth boundary, the coefficients $\omega$, $h\in L^1(\Omega)$ are nonnegative and $2<q<r$. They first showed that the existence, nonexistence and multiplicity results depend on $\lambda$ and the integrability of the ratio $w^{r-1}/h^{q-1}$. In \cite{PR}, {\em Pucci} and {\em R\u{a}dulescu} first considered the following related problem in the whole space:
\begin{eqnarray}\label{1.2}
-{\rm div}(|\nabla u|^{p-2}\nabla u)+u^{p-1}=\lambda u^{q-1}-h(x)u^{r-1}\ \ &{\rm in}\ \mathbb{R}^N,
\end{eqnarray}
where $h>0$ satisfies
\begin{align*}
0<\int_{\mathbb{R}^N}h(x)^{q/(q-r)}dx<\infty,
\end{align*}
$\lambda>0$ is a parameter and $2\leq p<q<\min\{r,p^*\}$, with $p^*=Np/(N-p)$ if $N>p$ and $p^*=\infty$ if $N\leq p$.
They obtained that the nonexistence and multiplicity of nontrivial solutions are corresponding to the smallness and largeness of $\lambda$ respectively. In \cite{r24}, {\em Autuori} and {\em Pucci} extended the above result by studying the quasilinear elliptic equation:
\begin{align}\label{1.3}
-{\rm div}(|\nabla u|^{p-2}\nabla u)+a(x)u^{p-2}u=\lambda \omega(x) u^{q-2}u-h(x)u^{r-2}u\ \ &{\rm in}\ \mathbb{R}^N,
\end{align}
where $\max\{2,p\}<q<\min\{r,p^*\}$, the coefficients $\omega$ and $h$ are related by integrability condition
\begin{align}\label{hw}
\int_{\mathbb{R}^N}\left[\frac{\omega^r(x)}{h^q(x)}\right]^{1/(r-q)}dx\in\mathbb R^+.
\end{align}
 Moreover, they proposed two open questions: the relaxation of $\max\{2,p\}<q$ and the replacement of \eqref{hw} by the assumption that $\omega\left(\omega/h\right)^{(q-p)/(r-q)}$ is in $L^{N/p}(\mathbb{R}^N)$. Note that thanks to $q<p^*$, this latter request is weaker than \eqref{hw} which already appeared in \cite{r36}. After that, {\em Autuori} and  {\em Pucci} in \cite{r35} turned to study the following elliptic equation involving fractional Laplacian
\begin{align}\label{1.7}
(-\Delta)^su+a(x)u=\lambda \omega(x)|u|^{p-2}u-h(x)|u|^{r-2}u\  \  {\rm in}\ \mathbb{R}^N,
\end{align}
where $\lambda\in\mathbb{R}$, $0<s<1$, $2s<N$, $2<q<\min\{r,2^*_s\}$, $2^*_s:=2N/(N-2s)$. The coefficients $\omega$ and $h$ are related by condition \eqref{hw}. By using variational methods, the authors obtained the existence and multiplicity of entire solutions of \eqref{1.7}. Similarly, two open problems mentioned above in \cite{r24} can be applied to the fractional setting. Recently, {\em Pucci} and {\em Zhang} \cite{PZ} solved the above open problems for a class of quasilinear elliptic equations in the setting of variable exponents. More recently, {\em Pucci et.al.} in \cite{PXZ2} also gave a positive answer to these open problems in the context of fractional $p$-Laplacian. In this article, we would like to take a quite different approach to conquer these problems in the same setting.

To this end, we suppose that the nonlinear terms in \eqref{k1} are related to the main elliptic part by the request that
{\em\begin{itemize}
\item[${\rm (H1)}$] $p<r<\min\{q,p_s^*\}$,
where $p_s^*=Np/(N-ps)$ is the fractional Sobolev critical exponent;
\item[${\rm (H2)}$] $V:\mathbb{R}^N\rightarrow(0,\infty)$\ is a positive weight and there  exists a constant $V_0>0$\ such that
$V(x)\geq V_0$\  for all $x\in \mathbb{R}^N$;
\item[${\rm (H3)}$] $a:\mathbb{R}^N\rightarrow(0,\infty)$\ is a positive weight satisfying  $a\in L^{\frac{p^*_s}{p^*_s-r}}(\mathbb{R}^N)\bigcap L^1_{\rm loc}(\mathbb{R}^N)$;
\item[${\rm (H4)}$] $b:\mathbb{R}^N\rightarrow(0,\infty)$\ is of class $L^1_{\rm loc}(\mathbb{R}^N)$, $a(x)$\  and\ $b(x)$\   are related by the condition
$$0<\int_{\mathbb{R}^N}\left[\frac{a(x)^{(q-p)/(q-r)}}{b(x)^{(r-p)/(q-r)}}\right]^{\frac{N}{ps}}dx<\infty,$$
i.e.,
$$ a\left(a/b\right)^{(r-p)/(q-r)}\in L^{N/ps}(\mathbb{R}^N).$$
\end{itemize}}










We first give the definition of weak solutions for problem \eqref{k1}.
\begin{definition}\label{definition1.1}
\emph{We say that $u\in W$ is a weak solution of problem \eqref{k1}, if
\begin{align*}
&\int_{\mathbb{R}^{2N}} \frac{|u(x)-u(y)|^{p-2}(u(x)-u(y))(\varphi(x)-\varphi(y))}{|x-y|^{N+ps}}dxdy+\int_{\mathbb{R}^N}V(x)|u(x)|^{p-2}u(x)\varphi(x) dx\\
&=\lambda \int_{\mathbb{R}^N}a(x)|u(x)|^{r-2}u(x)\varphi(x)dx-\int_{\mathbb{R}^N}b(x)|u(x)|^{q-2}u(x)\varphi(x) dx,
\end{align*}
for any $\varphi\in W$, where space $W$ will be introduced in section 2.}
\end{definition}
Now we are in a position to state the main result of this paper as follows.
\begin{theorem}\label{Th1}
Suppose that the assumptions {\em (H1)--(H4)} are satisfied. Then there exist constants $\lambda_0$ and $\lambda^*$ with $\lambda^*\geq\lambda_0>0$ such that problem \eqref{k1} has
\begin{itemize}
\item[$(\textrm{i})$] only the trivial weak solution if $\lambda<\lambda_0$;
\item[$(\textrm{ii})$] two nontrivial weak solutions if $\lambda>\lambda^*$.
\end{itemize}
\end{theorem}

\noindent\textbf{Remark 1.1.} (a) For the related results about concave convex problems, we refer to the seminal paper \cite{ABC}
for the semilinear case in bounded domains. We also refer to \cite{ XZG} for the fractional Laplacian case in bounded domains
and to \cite{XZF} for the fractional $p$-Laplacian case in $\mathbb{R}^N$.

 (b) Under the conditions (H1) and (H3), the condition (H4) is more general than that in \cite[(1.4)]{r35} for fractional Laplacian problems, see \cite{r36} for further details. Consequently, we obtain the corresponding results by replacing condition \cite[(1.4)]{r35} with (H4). Hence, in this sense our main result is new even in the case of the fractional Laplacian. Of course, from Theorem 1.1 we know that it still remains an open problem to verify whether $\lambda_0=\lambda^*$ in the non-local setting.


Finally, we would like to point out that in recent years, a great attention has been focused on the study of fractional and non-local operators of elliptic type. This type of operators arises in a quite natural way in many different applications, such as, continuum mechanics, phase transition phenomena, population dynamics
and game theory, as they are the typical outcome of stochastically stabilization of L\'{e}vy processes, see \cite{r2, r-1, L1, r-3}. The literature on fractional and non-local operators of elliptic and their applications is quite large, for example, we refer the reader to
\cite{r4, r1, CW, r-8, r-4, MR, r3, r17, r14}  and the references therein. For the basic properties of fractional Sobolev spaces with applications to partially differential equations, we refer the reader to \cite{r8, MRS} and the references therein.






This article is organized as follows. In Section 2, we will introduce the working space $W$ and give some necessary definitions and properties, which will be used in the sequel. In Section 3, using critical point theory, we will prove the main result.




\section{The Functional Framework}

In this section, we first give some basic results of fractional Sobolev space that will be used in the next section.
Let $0<s<1<p<\infty$ be real numbers.
The fractional Sobolev space $W^{s,p}(\mathbb{R}^N)$ is defined as follows
\begin{align*}
W^{s,p}(\mathbb{R}^N)=\left\{u\in L^{p}(\mathbb{R}^N): \int_{\mathbb{R}^{2N}}\frac{|u(x)-u(y)|^{p}}{|x-y|^{N+ps}}dxdy<\infty\right\},
\end{align*}
equipped  with the norm
$$\|u\|_{W^{s,p}(\mathbb{R}^N)}=\left(\|u\|_{L^p(\mathbb{R}^N)}^p
+\int_{\mathbb{R}^{2N}}\frac{|u(x)-u(y)|^{p}}{|x-y|^{N+ps}}dxdy\right)^{\frac{1}{p}}.$$
Then it is easy to see that $(W^{s,p}(\mathbb{R}^N), \|u\|_{W^{s,p}(\mathbb{R}^N)})$ is an uniformly convex Banach space (see and the embedding $W^{s,p}(\mathbb{R}^N)\hookrightarrow L^{\nu}(\mathbb{R}^N)$ is continuous for any $\nu\in [p,p_s^*]$ by Theorem 6.7 of \cite{r8}, that is, there exists a positive constant $C_*$ such that
\begin{align}\label{embedding}
\|u\|_{L^{\nu}(\mathbb{R}^N)}\leq C_* \|u\|_{W^{s,p}(\mathbb{R}^N)}\ \ {\rm for\ all}\ u\in W^{s,p}(\mathbb{R}^N).
\end{align}
The space $E$ denotes the completion of $C_0^{\infty}(\mathbb{R}^N)$ with respect to the norm
\begin{align*}
\|u\|_{E}=\left(\int_{\mathbb{R}^{2N}}\frac{|u(x)-u(y)|^{p}}{|x-y|^{N+ps}}dxdy+\int_{\mathbb{R}^N}V(x)|u(x)|^{p}dx\right)^{\frac{1}{p}}.
\end{align*}
Then $(E.\|\cdot\|_E)$ is a uniformly convex Banach space, see \cite[Lemma 10]{PXZ1}.
\begin{lemma}\label{lemma2.1} {\em (see \cite[Lemma 1]{PXZ1})}
Let {\em (H2)} holds. Then the embeddings $E\hookrightarrow W^{s,p}(\mathbb{R}^N)\hookrightarrow L^{\nu}(\mathbb{R}^N)$ are continuous with
\begin{equation}\label{embedding2}
\min\{1,V_0\}\|u\|^p_{W^{s,p}(\mathbb{R}^N)}\leq \|u\|^p_{E},
\end{equation}
for all $u\in W$ and $\nu\in [p,p_s^*]$. Moreover, for any $R>0$ and $\nu\in [1,p_s^*)$ the embedding $E\hookrightarrow L^\nu(B_R(0))$ is compact.
\end{lemma}



$L^r(\mathbb{R}^N,a)$ is defined by a linear space of Lebesgue measurable functions from $\mathbb{R}^N$ to $\mathbb{R}$ such that for any function $u$ in $L^q(\mathbb{R}^N,a)$
$$\int_{\mathbb{R}^N}a(x)|u(x)|^rdx<\infty,$$
equipped with the norm
\begin{align*}
\|u\|_{L^r(\mathbb{R}^N,a)}=\left(\int_{\mathbb{R}^N}a(x)|u(x)|^{r}dx\right)^{\frac{1}{r}},
\end{align*}
then $L^r(\mathbb{R}^N,a)$ is an uniformly convex Banach space by Proposition A.6 of \cite{r24}. The space $L^q(\mathbb{R}^N, b)$ is defined in a similar way.
We shall work on the space $W$ defined as follows:
$W$ is the completion of $C_0^\infty(\mathbb{R}^N)$ with respect to the norm
\begin{align*}
\|u\|_{W}=\|u\|_{E}+\|u\|_{L^q(\mathbb{R}^N,b)}.
\end{align*}
\begin{lemma}\label{lemma2.2}
$(W, \|\cdot\|_{W})$ is a reflexive Banach space.
\end{lemma}

\begin{proof}
We only need to verify that $(W, \|\cdot\|_{W})$ is reflexive. For this, we define a product space $X=E\times L^q(\mathbb{R}^N,b)$, endowed with the norm $$\|u\|_{X}=\|u\|_{E}+\|u\|_{L^q(\mathbb{R}^N,b)}.$$ Then $(X, \|\cdot\|_{X})$ is a reflexive Banach space by Theorem 1.23 (ii) of \cite{r22}, since $E$ and $L^q(\mathbb{R}^N,\ b)$ are two uniformly convex Banach spaces.
We define an operator $\Gamma: (W,\|\cdot\|_{W})\rightarrow (X,\|\cdot\|_{X})$ satisfying $\Gamma (u)=(u,u)$ for all $u\in W$. Then $\Gamma$ is well defined, linear and isometric. Therefore, $\Gamma(W)$ is a closed subspace of $X$, and so  $\Gamma(W)$ is reflexive by Theorem 1.22 (ii) of \cite{r22}. Consequently, $(W,\|\cdot\|_{W})$ is reflexive.
\end{proof}


From now on, $B_R(0)$ will denote the ball in $\mathbb{R}^N$ of center zero and radius $R>0$.
\begin{theorem}\label{th4}
Suppose that $V:\mathbb{R}^N\rightarrow(0,\infty)$ satisfies assumption {\em (H2)}, $a:\mathbb{R}^N\rightarrow(0,\infty)$ satisfies assumption
 {\em (H3)} and $b:\mathbb{R}^N\rightarrow(0,\infty)$ is a positive weight. Let $\{v_j\}$ be a bounded sequence in $W$, then there exists $v\in W\bigcap L^{r}(\mathbb{R}^N)$ such that up to a subsequence,
\begin{align*}
v_j\rightarrow v\ \ {\rm strongly\ in}\ L^r(\mathbb{R}^N,a)
\end{align*}
as $j\rightarrow\infty$, for any $r\in(p,p_s^*)$.
\end{theorem}
\begin{proof}
Since $\{v_j\}_{j}$ is bounded in $W$, by Lemma \ref{lemma2.1} we have $\{v_j\}_{j}$ is bounded in $L^{p_s^*}(\mathbb{R}^N)$. Then by the reflexivity of $W$, up to a subsequence, we get that $v_j\rightharpoonup v$ weakly in $W\bigcap L^{p_s^*}(\mathbb{R}^N)$ as $j\rightarrow\infty$. Next we prove that
\begin{align*}
v_j\rightarrow v\ \ {\rm strongly\ in}\ L^{r}(\mathbb{R}^N,a).
\end{align*}
 Now for any $\varepsilon>0$, there exists $R_1>0$ such that
\begin{align}\label{theorem2.1.1}
\int_{\mathbb{R}^N\backslash B_R(0)}|a(x)|^{\frac{p_s^*}{p_s^*-r}}dx<\varepsilon \ \ {\rm for\ all}\ R\geq R_1,
\end{align}
since $a\in L^{\frac{p_s^*}{p_s^*-r}}(\mathbb{R}^N)$ by assumption (H3).
Fix $R_1>0$, we have $$v_j\rightharpoonup v\ \ {\rm weakly\ in}\ \ W^{s,p}(B_{R_1}(0))\bigcap L^{p_s^*}(B_R(0)),$$ by Theorem 6.7 of \cite{r8}. Since $p<r<p_s^*$, by Corollary 7.2 of \cite{r8}, we obtain $v_j\rightarrow v$ strongly in $L^r(B_{R_1}(0))$. Without loss of generality, we assume that $u_n\rightarrow u$ a.e. in $B_{R_1}(0)$. Thus,  $a(x)|v_j-v|^r\rightarrow 0$ a.e. in $B_{R_1}(0)$. For each measurable subset $U\subset B_{R_1}(0)$, we have
\begin{align*}
\int_{U}a(x)|v_j(x)-v(x)|^r dx\leq \|a\|_{L^{\frac{p_s^*}{{p_s^*-r}}}(U)}\|v_j-v\|^{r}_{L^{p_s^*}(U)}
\leq C\|a\|_{L^{\frac{p_s^*}{{p_s^*-r}}}(U)}.
\end{align*}
 Since $a\in L^{\frac{p_s^*}{p_s^*-r}}(\mathbb{R}^N)$, we obtain that $\{a(x)|v_j(x)-v(x)|^r\}_{j}$ is equi--integrable and uniformly bounded in $B_{R_1}(0)$. Then the Vitali convergence theorem implies
\begin{align*}
\lim_{j\rightarrow\infty}\int_{B_{R_1(0)}}a(x)|v_j(x)-v(x)|^r dx=0.
\end{align*}
Hence, for above $\varepsilon>0$, there exists $N_1>0$ such that
\begin{align}\label{theorem2.1.2}
\int_{B_{R_1(0)}}a(x)|v_j(x)-v(x)|^r dx
<\varepsilon,\ \ {\rm for\ all}\ j\geq N_1.
\end{align}
Combining \eqref{theorem2.1.1} and \eqref{theorem2.1.2}, for all $j\geq N_1$ we have
\begin{align*}
\int_{\mathbb{R}^N}a(x)|v_j(x)-v(x)|^rdx&=\int_{B_{R_1(0)}}a(x)|v_j(x)-v(x)|^rdx+\int_{\mathbb{R}^N\backslash B_{R_1(0)}}a(x)|v_j(x)-v(x)|^rdx\\
&<\varepsilon+\left(\int_{\mathbb{R}^N\backslash B_{R_1(0)}}|a(x)|^{\frac{p_s^*}{p_s^*-r}}\right)^{\frac{r}{q}}\left(\int_{\mathbb{R}^N\backslash B_{R_1(0)}}|v_j(x)-v(x)|^{p_s^*}dx\right)^{\frac{r}{p_s^*}}\\
&\leq C\varepsilon,
\end{align*}
where $C$ denotes various positive constants.
Therefore, $v_j\rightarrow v$ strongly in $L^r(\mathbb{R}^N,a)$.
\end{proof}


\section{Proof of Theorem 1.1}

For $u\in W$, we define
\begin{align*}
&J(u)=\frac{1}{p}\int_{\mathbb{R}^{2N}}\frac{|u(x)-u(y)|^{p}}{|x-y|^{N+ps}}dxdy
+\frac{1}{p}\int_{\mathbb{R}^N}V(x)|u(x)|^pdx+\frac{1}{q}\int_{\mathbb{R}^N}b(x)|u(x)|^qdx,\\ &H(u)=\frac{\lambda}{r}\int_{\mathbb{R}^N}a(x)|u(x)|^rdx,
\end{align*}
and
$$I(u)=J(u)-H(u).$$
Obviously, the energy functional $I:W\rightarrow\mathbb{R}$ associated with problem \eqref{k1} is well defined.

\begin{lemma}\label{lemma3.1}
Under the conditions {\em (H1)--(H4)}, the functional $I$ is coercive in $W$.
\end{lemma}
\begin{proof}
For any $k_1,k_2>0$ and $0<\alpha<\beta$
\begin{align}\label{elementary}
k_1t^\alpha-k_2t^\beta\leq k_1\left(k_1/k_2\right)^{\alpha/(\beta-\alpha)}\ \ {\rm for\ all}\ t\geq0.
\end{align}
Indeed, since the function $t\in[0,\infty)\rightarrow t^{\theta}$ is increasing for any $\theta>0$, it follows that
 $$k_1-k_2t^{\beta-\alpha}<0,\ \forall t>\left(k_1/k_2\right)^{1/(\beta-\alpha)}$$
 and
 $$k_1t^\alpha-k_2t^\beta\leq k_1t^\alpha\leq k_1 \left(k_1/k_2\right)^{\alpha/(\beta-\alpha)},\ \ \forall t\in\left[0,\ \left(k_1/k_2\right)^{1/(\beta-\alpha)}\right]. $$
The above two inequality imply that \eqref{elementary} holds.



Taking $k_1=\lambda a(x)/r, k_2=b(x)/q, \alpha=r-p,\beta=q-p$ and $t=|u(x)|$ in \eqref{elementary}, for all $x\in\mathbb{R}^N$, we get
\begin{align}\label{elementary2}
\frac{\lambda a(x)}{r}|u(x)|^{r-p}-\frac{b(x)}{2q}|u(x)|^{q-p}
\leq(\lambda/r)^{(q-p)/(q-r)}(2q)^{(r-p)/(q-r)}\left(\frac{a(x)^{q-p}}{b(x)^{r-p}}\right)^{1/(q-r)}.
\end{align}
Since $\left(a^{q-p}/b^{r-p}\right)^{1/(q-r)}\in L^{N/(ps)}(\mathbb{R}^N)$, for any $\varepsilon>0$ there exists $R>0$ such that
\begin{align}\label{elementary3}
(\lambda/r)^{(q-p)/(q-r)}(2q)^{(r-p)/(q-r)}\left(\int_{\mathbb{R}^N\backslash B_R(0)}\left(\frac{a(x)^{q-p}}{b(x)^{r-p}}\right)^{N/ps(q-r)}dx\right)^{ps/N}<\frac{\min\{1,V_0\}}{2C_*^p}\varepsilon.
\end{align}
So, by \eqref{elementary2} and \eqref{elementary3} we obtain
\begin{align}\label{elementary33}
I(u)=&\frac{1}{p}\int_{\mathbb{R}^{2N}}\frac{|u(x)-u(y)|^{p}}{|x-y|^{N+ps}}dxdy
+\frac{1}{p}\int_{\mathbb{R}^N}V(x)|u(x)|^pdx+\frac{1}{q}\int_{\mathbb{R}^N}b(x)|u(x)|^qdx-\frac{\lambda}{r}\int_{\mathbb{R}^N}a(x) |u(x)|^rdx\nonumber\\
\geq& \frac{1}{p}\int_{\mathbb{R}^{2N}}\frac{|u(x)-u(y)|^{p}}{|x-y|^{N+ps}}dxdy
+\frac{1}{p}\int_{\mathbb{R}^N}V(x)|u(x)|^pdx
+\frac{1}{2q}\int_{\mathbb{R}^N}b(x)|u(x)|^qdx\nonumber\\
&-(\lambda/r)^{(q-p)/(q-r)}(2q)^{(r-p)/(q-r)}\int_{\mathbb{R}^N\backslash B_R(0)}\left(\frac{a(x)^{q-p}}{b(x)^{r-p}}\right)^{1/(q-r)}|u(x)|^pdx\nonumber\\
&+\int_{B_R(0)}\frac{b(x)}{2q}|u(x)|^q-\frac{\lambda a(x)}{r} |u(x)|^rdx.
\end{align}
For fixed $R>0$ and for any $\delta>0$ and $M>0$, decompose $B_R(0)=X\bigcup Y\bigcup Z$ with $X,Y,Z$ measurable sets defined as follows:
\begin{align}\label{elementary5}
\begin{cases}
X=\{x\in B_R(0): a(x)<M \ \ {\rm and}\ \ b(x)>\delta\},\\
Y=\{x\in B_R(0): a(x)<M\ \ {\rm and}\ \ b(x)\leq\delta\},\\
Z=\{x\in B_R(0): a(x)\geq M\}.
\end{cases}
\end{align}
We apply \eqref{elementary} to derive
\begin{align}\label{elementary6}
&\int_{X}\frac{\lambda a(x)}{r} |u(x)|^r-\frac{b(x)}{2q}|u(x)|^qdx\nonumber\\
&\leq\left(\lambda/r\right)^{q(q-r)}(2q)^{r/(q-r)}\int_{X}\frac{a(x)^{q/(q-r)}}{b(x)^{r/(q-r)}}dx\leq C_1,
\end{align}
and
\begin{align}\label{elementary7}
&\int_{Y\bigcup Z}\frac{\lambda a(x)}{r} |u(x)|^r-\frac{b(x)}{2q}|u(x)|^qdx\nonumber\\
&\leq(\lambda/r)^{(q-p)/(q-r)}(2q)^{(r-p)/(q-r)}\int_{Y\bigcup Z}\frac{a(x)^{(q-p)/(q-r)}}{b(x)^{(r-p)/(q-r)}} |u(x)|^pdx\nonumber\\
&\leq (\lambda/r)^{(q-p)/(q-r)}(2q)^{(r-p)/(q-r)}\left(\int_{Y\bigcup Z}\left[\frac{a(x)^{(q-p)/(q-r)}}{b(x)^{(r-p)/(q-r)}}\right]^{N/ps}dx\right)^{ps/N}\|u\|^p_{L^{p_s^*}(B_R(0))},
\end{align}
where $C_1=C_1(M,\delta, R)$ is a constant. Since $a, b\in L^1_{\rm loc}(\mathbb{R}^N)$, we have $|Z|\rightarrow0$ as $M\rightarrow\infty$ and for fixed $M$, $|Y|\rightarrow0$ as $\delta\rightarrow0$. Thus, we can choose $M$ sufficiently large and then $\delta>0$ sufficiently small such that
\begin{align}\label{elementary8}
(\lambda/r)^{(q-p)/(q-r)}(2q)^{(r-p)/(q-r)}\left(\int_{Y\bigcup Z}\left[\frac{a(x)^{(q-p)/(q-r)}}{b(x)^{(r-p)/(q-r)}}\right]^{N/ps}dx\right)^{ps/N}
< \frac{\min\{1,V_0\}}{2 C_*^p}\varepsilon,
\end{align}
where $C_*$ denotes the embedding constant of $W^{s,p}(\mathbb{R}^N)\hookrightarrow L^{p}(\mathbb{R}^N)$.
Combining \eqref{elementary6}, \eqref{elementary7} and  \eqref{elementary8}, we conclude
\begin{align}\label{elementary9}
\int_{B_R(0)}\frac{\lambda a(x)}{r} |u(x)|^r-\frac{b(x)}{2q}|u(x)|^qdx\leq C_1+ \frac{\varepsilon}{2}\|u\|_{E}^p
\end{align}
By \eqref{elementary33} and \eqref{elementary9}, we obtain
\begin{align*}
I(u)\geq& \frac{1}{p}\int_{\mathbb{R}^{2N}}\frac{|u(x)-u(y)|^{p}}{|x-y|^{N+ps}}dxdy
+\frac{1}{p}\int_{\mathbb{R}^N}V(x)|u(x)|^pdx
+\frac{1}{2q}\int_{\mathbb{R}^N}b(x)|u(x)|^qdx\nonumber\\
&-\varepsilon \|u\|^p_{E}-C_1
\end{align*}
Taking $\displaystyle\varepsilon=1/2p$ and using the following inequality
\begin{align*}
\xi^{l}\geq \xi-1 \ \ {\rm for\ all}\ \xi\geq0\ {\rm and}\ l\geq1,
\end{align*}
we get
\begin{align*}
I(u)&\geq \frac{1}{2p}\int_{\mathbb{R}^{2N}}\frac{|u(x)-u(y)|^{p}}{|x-y|^{N+ps}}dxdy
+\frac{1}{2p}\int_{\mathbb{R}^N}V(x)|u(x)|^pdx
+\frac{1}{2q}\int_{\mathbb{R}^N}b(x)|u(x)|^qdx-C_1\\
&\geq \min\left\{\frac{1}{2p},\frac{1}{2q}\right\}\|u\|_W-\frac{1}{2p}-\frac{1}{2q}-C_1.
\end{align*}
Hence, $I$ is coercive in $W$.
\end{proof}

\begin{lemma}\label{lemma3.2}
The functional $J: W\rightarrow\mathbb{R}$ is convex and of class $ C^1$ and
\begin{align*}
\langle J^\prime(u),v\rangle&=\int_{\mathbb{R}^{2N}}\frac{ |u(x)-u(y)|^{p-2}(u(x)-u(y))(v(x)-v(y))}{|x-y|^{N+ps}}dxdy\\
&+\int_{\mathbb{R}^N}V(x)|u(x)|^{p-2}u(x)v(x)dx+\int_{\mathbb{R}^N}b(x)|u(x)|^{q-2}u(x)v(x)dx,
\end{align*}
for all $\ u,\ v\in W$. Moreover, $J$ is weakly lower semi-continuous in $W$.
\end{lemma}
\begin{proof}
It is easy to see that $\psi$ is G\^{a}teaux-differentiable in $W$ and for all $u,v\in W$
\begin{align*}
\langle J^\prime(u),v\rangle&=\int_{\mathbb{R}^{2N}}\frac{ |u(x)-u(y)|^{p-2}(u(x)-u(y))(v(x)-v(y))}{|x-y|^{N+ps}}dxdy\\
&+\int_{\mathbb{R}^N}V(x)|u(x)|^{p-2}u(x)v(x)dx+\int_{\mathbb{R}^N}b(x)|u(x)|^{q-2}u(x)v(x)dx,
\end{align*}
Now, let $\{u_n\}\subset W, u\in W$ satisfy $u_n\rightarrow u$ strongly in $W$ as $n\rightarrow\infty$. Without loss of generality, we assume that $u_n\rightarrow u$ a.e. in $\mathbb{R}^N$. Then the sequence
\begin{align*}
\left\{\frac{ |u_n(x)-u_n(y)|^{p-2}(u_n(x)-u_n(y))}{|x-y|^{(N+ps)/p^\prime}} \right\}_{n} \ {\rm\ is\ bounded\ in}\ L^{p^\prime}(\mathbb{R}^{2N})
\end{align*}
as well as
\begin{align*}
\frac{ |u_n(x)-u_n(y)|^{p-2}(u_n(x)-u_n(y))}{|x-y|^{(N+ps)/p^\prime}}\longrightarrow\frac{ |u(x)-u(y)|^{p-2}(u(x)-u(y))}{|x-y|^{(N+ps)/p^\prime}}  \ \ {\rm\ a.\ e.\ in}\ \mathbb{R}^{2N},
\end{align*}
as $n\rightarrow \infty$. Thus, the Brezis--Lieb lemma (see \cite{r21}) implies
\begin{align}\label{three1}
&\lim_{n\rightarrow\infty}\int_{\mathbb{R}^{2N}}\left|\frac{ |u_n(x)-u_n(y)|^{p-2}(u_n(x)-u_n(y))-|u(x)-u(y)|^{p-2}(u_n(x)-u_n(y))}{|x-y|^{(N+ps)/p^\prime}}\right|^{p^\prime}dxdy\nonumber\\
&=\lim_{n\rightarrow\infty}\int_{\mathbb{R}^{2N}}\left(\frac{ |u_n(x)-u_n(y)|^{p}}{|x-y|^{N+ps}}-\frac{ |u(x)-u(y)|^{p}}{|x-y|^{N+ps}}\right)dxdy.
\end{align}
Since $u_n\rightarrow u$ strongly in $W$, it is easy to see that
$$\lim_{n\rightarrow\infty}\int_{\mathbb{R}^{2N}}\left(\frac{ |u_n(x)-u_n(y)|^{p}}{|x-y|^{N+ps}}-\frac{ |u(x)-u(y)|^{p}}{|x-y|^{N+ps}}\right)dxdy=0.$$
It follows from \eqref{three1} that
\begin{align}\label{three2}
&\lim_{n\rightarrow\infty}\int_{\mathbb{R}^{2N}}\left|\frac{ |u_n(x)-u_n(y)|^{p-2}(u_n(x)-u_n(y))-|u(x)-u(y)|^{p-2}(u_n(x)-u_n(y))}{|x-y|^{(N+ps)/p^\prime}}\right|^{p^\prime}dxdy\nonumber\\
&=0.
\end{align}
Similarly, we obtain
\begin{align}\label{three3}
 \lim_{n\rightarrow\infty}\int_{\mathbb{R}^N} V(x)\left||u_n(x)|^{p-2}u_n(x)-|u(x)|^{p-2}u(x)\right|^{p^\prime}dx=0.
\end{align}
and
\begin{align}\label{three4}
 \lim_{n\rightarrow\infty}\int_{\mathbb{R}^N}b(x)\left||u_n(x)|^{q-2}u_n(x)-|u(x)|^{q-2}u(x)\right|^{q^\prime}dx=0.
\end{align}
Combining \eqref{three2}--\eqref{three4} with the H\"{o}lder inequality, we have
\begin{align*}
\|J^\prime(u_n)-J^\prime(u)\|_{W^\prime}&=\sup_{\varphi\in W,\ \|\varphi\|_{W}=1}|\langle J^\prime(u_n)-J^\prime(u),\varphi\rangle|\\
&\longrightarrow0,\ \ {\rm as}\ n\rightarrow\infty.
\end{align*}
Hence, $J\in C^1(W,\mathbb{R})$. Obviously, $J$ is a convex functional in $W$. Furthermore, $J$ is weakly lower semi-continuous in $W$ by Corollary 3.9 of \cite{r24}.
\end{proof}


\begin{lemma}\label{lemma3.3}
$H\in C^1(W,\mathbb{R})$ and
for any fixed $u\in W$, we have
\begin{align*}
\langle H^\prime(u(x)),v(x)\rangle=\lambda\int_{\mathbb{R}^N}a(x)|u(x)|^{r-2}u(x)v(x)dx,
\end{align*}
and $H^\prime(u)\in W^\prime$. If $v_n\rightharpoonup v$ weakly in $W$, then $\langle H^\prime(u),v_n\rangle\rightarrow\langle H^\prime(u),v\rangle$.
Moreover, under the condition {\em (H1)}, functional $H$ is weakly continuous in $W$.
\end{lemma}
\begin{proof}
We only need to prove that $H$ is weakly continuous in $W$ and is of class $ C^1$. Let $\{u_n\}\subset W, u\in W$ satisfy $u_n\rightharpoonup u$ weakly in $W$ as $n\rightarrow\infty$. By Theorem \ref{th4},  without loss of generality, we assume that $u_n\rightarrow u$ strongly in $L^r(\mathbb{R}^N,a)$ for $p<r<p_s^*$ and a.e. in $\mathbb{R}^N$. Then a similar argument as in Lemma \ref{lemma3.2} gives that
\begin{align*}
\lim_{n\rightarrow\infty}\int_{\mathbb{R}^N}a(x)|u_n(x)|^{r}dx=\int_{\mathbb{R}^N}a(x)|u_n(x)|^{r}dx
\end{align*}
and
\begin{align*}
\lim_{n\rightarrow\infty}\int_{\mathbb{R}^N}a(x)\left||u_n(x)|^{r-2}u_n(x)-|u(x)|^{r-2}u(x)\right|^{r^\prime}dx=0.
\end{align*}
So it is easy to verify that $H$ is weakly continuous in $W$ and is of class $ C^1$.
\end{proof}

Combining Lemma \ref{lemma3.2} and Lemma \ref{lemma3.3}, we get that $I\in C^1(W,\mathbb{R})$ and $I$ is weakly semi-continuous in $W$. To solve problem \eqref{k1}, we first consider
\begin{align*}\lambda^*:&=\inf_{u\in\mathcal{M}}\left(\frac{r}{p}\int_{\mathbb{R}^{2N}}\frac{|u(x)-u(y)|^{p}}{|x-y|^{N+ps}}dxdy+\frac{r}{p}\int_{\mathbb{R}^N}V(x)|u(x)|^pdx+\frac{r}{q}\int_{\mathbb{R}^N}b(x)|u(x)|^qdx\right)\\
&=\inf_{u\in\mathcal{M}}J(u),
\end{align*}
where $\mathcal{M}=\left\{u\in W: \int_{\mathbb{R}^N}a(x)|u(x)|^rdx=1\right\}$.

\begin{lemma}\label{lemma3.4}
$\inf\limits_{u\in\mathcal{M}} J(u)$ is achieved at some $u_0\in \mathcal{M}$ and $\lambda^*:=\inf\limits_{u\in\mathcal{M}}J(u)=J(u_0)>0$. Particularly, $J(|u_0|)=J(u_0)$, that is, $|u_0|$ is also a global minimizer of $\inf\limits_{u\in\mathcal{M}} J(u)$.
\end{lemma}
\begin{proof}
Let $\{u_n\}$ be a minimizing sequence. Then
\begin{align*}
J(u_n)\rightarrow \inf_{\mathcal{M}}J\ \ {\rm as}\ n\rightarrow\infty.
\end{align*}
Hence $\{u_n\}$ is bounded in $W$. By Theorem \ref{th4}, $\{u_n\}$ has a convergent subsequence in $L^r(\mathbb{R}^N, a)$, i. e. up to a subsequence , $u_n\rightarrow u_0$ strongly in $L^r(\mathbb{R}^N,a)$. Then we have
\begin{align*}
1=\int_{\mathbb{R}^N}a(x)|u_n(x)|^rdx\rightarrow\int_{\mathbb{R}^N}a(x)|u_0(x)|^rdx \ \ {\rm as}\ n\rightarrow\infty.
\end{align*}
Thus, $u_0\in \mathcal{M}$. Since $J$ is weakly lower semi-continuous, we have
\begin{align*}
\inf_{\mathcal{M}}J\leq J(u_0)\leq \lim_{n\rightarrow\infty} J(u_n)=\inf_{\mathcal{M}}J.
\end{align*}
Therefore, $J(u_0)=\inf\limits_{\mathcal{M}}J$. If $\lambda^*=0$, then
\begin{align*}
J(u_n)\rightarrow0\ \ {\rm as}\ n\rightarrow\infty.
\end{align*}
Thus
\begin{align*}
\int_{\mathbb{R}^N}a(x)|u_n(x)|^rdx\rightarrow0\ \ {\rm as}\ n\rightarrow\infty,
\end{align*}
which contradicts with the fact $u_n\in\mathcal{M}$. So, $\lambda^*>0$. \\
Finally, we show that $J(|u_0|)=J(u_0)$. For $v\in W$, we have $|v|\in W$. Indeed, we have
\begin{align*}
&\int_{\mathbb{R}^{2N}}\frac{||v(x)|-|v(y)||^p}{|x-y|^{N+ps}}dxdy\leq \int_{\mathbb{R}^{2N}}\frac{|v(x)-v(y)|^p}{|x-y|^{N+ps}}dxdy.
\end{align*}
Thus, it is easy to obtain that $|v|\in W$. Further, $|u_0|\in W$ and $J(|u_0|)\leq J(u_0)$. This gives $J(|u_0|)=J(u_0)$, due to the minimality of $u_0$.
\qed
\begin{theorem}\label{th3.1}
For all $\lambda>\lambda^*$, there exists a global nontrivial minimizer $u^*\in W$ of $I$ with $I(u^*)<0.$
\end{theorem}
{\it Proof.}
For all $\lambda>0$ the functional $I$ is weakly semi-continuous, bounded below and coercive in the reflexive Banach space $W$ by Lemma \ref{lemma3.1}--Lemma \ref{lemma3.3} and Lemma \ref{lemma2.1}. Hence, Theorem 1.2 of \cite{r23} implies that for all $\lambda>0$ there exists a global minimizer $u^*\in W$ of $I$, that is
$$I(u^*)=\inf_{u\in W}I(u).$$
Obviously, $u^*$ is a weak solution of \eqref{k1}.
Next, we show that $u^*\neq 0$. Since $\lambda>\lambda^*$, by Lemma \ref{lemma3.4} there exists a function $u_0\in W$ with $\|u_0\|_{L^r(\mathbb{R}^N,a)}=1$ such that
\begin{align*}\lambda \|u_0\|^r_{L^r(\mathbb{R}^N,a)}=\lambda>\lambda^*&=J(u_0)=\frac{r}{p}\mathop{\int}_{\mathbb{R}^{2N}}\frac{ |u_0(x)-u_0(y)|^{p}}{|x-y|^{N+ps}}dxdy\\&+\frac{r}{p}\int_{\mathbb{R}^N}V(x)|u_0(x)|^{p}dx+\frac{r}{q}\int_{\mathbb{R}^N}b(x)|u_0(x)|^{q}dx.
\end{align*}
This means that
\begin{align*}
I(u_0)&=\frac{1}{p}\int_{\mathbb{R}^{2N}}\frac{|u_0(x)-u_0(y)|^{p}}{|x-y|^{N+ps}}dxdy+\frac{1}{p}\int_{\mathbb{R}^N}V(x)|u_0(x)|^{p}dx\\
&+\frac{1}{q}\int_{\mathbb{R}^N}b(x)|u_0(x)|^{q}dx-\frac{\lambda}{r}\int_{\mathbb{R}^N}a(x)|u_0|^{r}dx<0.
\end{align*}
Consequently, $I(u^*)=\inf\limits_{u\in W}I(u)\leq I(u_0)<0$.
\end{proof}


Next, we show that if $\lambda>\lambda^*$ problem \eqref{k1} admits a second nontrivial weak solution $e\neq u$ via variational  methods. We start by recalling a modification of the mountain pass theorem of Ambrosetti and  Rabinowitz, see Theorem A.3 of \cite{r24}.
\begin{theorem}\label{th3.2}
Let $(X,\|\cdot\|_X)$ and $(Y,\|\cdot\|_{Y})$ be two Banach spaces such that $X$ can be continuously embedded into $Y$. Let $\Phi: X\rightarrow\mathbb{R}$ be a $C^1$ functional with $\Psi(0)=0$. Suppose that there exist $\rho,\alpha>0$ and $e\in X$ such that $\|e\|_{Y}>\rho, \Phi(e)<\alpha$ and $\Phi(u)\geq\alpha$ for all $u\in X$ with $\|u\|_{Y}=\rho$. Then there exists a sequence $\{u_n\}\subset X$ such that for all $n$
\begin{align*}
c\leq \Phi(u_n)\leq c+\frac{1}{n}\ \ {\rm and}\ \ \|\Phi^\prime(u_n)\|_{X^\prime}\leq \frac{2}{n},
\end{align*}
where
\begin{align*}
c=\inf_{\gamma\in\Gamma}\max_{t\in[0,1]}\Phi(\gamma(t))\ \ {\rm and}\ \ \Gamma=\{\gamma\in C([0,1];X): \gamma(0)=0,\gamma(1)=e\}.
\end{align*}
\end{theorem}
\begin{lemma}\label{lemma3.5}
Suppose that assumptions {\em (H1)--(H4)} are satisfied. Then for each $e\in W\backslash\{0\}$ and $\lambda>0$ there exist $\rho\in(0,\|e\|_{E})$ and $\alpha>0$ such that
 $$I(u)\geq\alpha>0,$$
for all $u\in W$ with $\|u\|_{E}=\rho$.
\end{lemma}
\begin{proof}
Let $u\in W$. By \eqref{embedding}, $a\in L^{\frac{p_s^*}{p_s^*-r}}(\mathbb{R}^N)$ and the H\"{o}lder inequality, we obtain
\begin{align*}
\int_{\mathbb{R}^N}a(x)|u(x)|^{r}dx&\leq \|a\|_{L^{\frac{p_s^*}{p_s^*-r}}(\mathbb{R}^N)}\|u\|_{L^{p_s^*}(\mathbb{R}^N)}^r\\
&\leq C_*^r\|a\|_{L^{\frac{p_s^*}{p_s^*-r}}(\mathbb{R}^N)}\|u\|_{W^{s,p}(\mathbb{R}^N)}^r\\
&\leq C_*^r\|a\|_{L^{\frac{p_s^*}{p_s^*-r}}(\mathbb{R}^N)}\|u\|_{W^{s,p}(\mathbb{R}^N)}^r.
\end{align*}
Then
\begin{align*}
I(u)&\geq\frac{1}{p}\int_{\mathbb{R}^{2N}}\frac{ |u(x)-u(y)|^{p}}{|x-y|^{N+ps}}dxdy+\frac{1}{p}\int_{\mathbb{R}^N}V(x)|u(x)|^{p}dx-\lambda C_*^r\|a\|_{L^{\frac{p_s^*}{p_s^*-r}}(\mathbb{R}^N)}\|u\|_{W^{s,p}(\mathbb{R}^N)}^r\\
&\geq \frac{1}{p}\|u\|_{E}^p-\lambda C_*^r\|a\|_{L^{\frac{p_s^*}{p_s^*-r}}(\mathbb{R}^N)}\|u\|_{W^{s,p}(\mathbb{R}^N)}^r\\
&\geq\|u\|_{E}^p\left(\frac{1}{p}-\lambda \left(\frac{C^p_*}{\min\{1,V_0\}}\right)^{r/p}\|a\|_{L^{\frac{p_s^*}{p_s^*-r}}(\mathbb{R}^N)}\|u\|_{E}^{r-p}\right).
\end{align*}
Now, let $\|u\|_{W^{s,p}(\mathbb{R}^N)}=\rho>0$ and $0<\rho<\left\{\|e\|_{E},\ \left(\lambda p \left(\frac{C^p_*}{\min\{1,V_0\}}\right)^{r/p}\|a\|_{L^{\frac{p_s^*}{p_s^*-r}}(\mathbb{R}^N)}\right)^{1/(p-r)}\right\}$, so that
\begin{align*}
I(u)\geq \rho^p\left(\frac{1}{p}-\lambda \left(\frac{C^p_*}{\min\{1,V_0\}}\right)^{r/p}\|a\|_{L^{\frac{p_s^*}{p_s^*-r}}(\mathbb{R}^N)}\rho^{r-p}\right)=:\alpha>0.
\end{align*}
Thus, the lemma is proved.
\end{proof}

\begin{theorem}\label{th3.3}
Suppose that assumptions {\em (H1)--(H4)} are satisfied. Then for all $\lambda>\lambda^*$ problem \eqref{k1} exists a nontrivial weak solution $u\in W$ such that $I(u)>0$.
\end{theorem}
\begin{proof}
By Theorem \ref{th3.1}, for all $\lambda>\lambda^*$ there exists a nontrivial weak solution $u^*\in W$, which is a global minimizer for $I$ in $W$ and $I(u^*)<0$. Taking $e=u^*$ in Lemma \ref{lemma3.5}, we know that $I$ satisfies the geometrical structure of Theorem \ref{th3.2}. Thus, for all $\lambda>\lambda^*$ there exists a sequence $\{u_n\}\subset W$ such that
\begin{align*}
I(u_n)\rightarrow c\ \ {\rm and}\ \ \|I^\prime(u_n)\|_{W^\prime}\rightarrow0\ \ {\rm as}\ n\rightarrow\infty,
\end{align*}
where
\begin{align*}
c=\inf_{\gamma\in\Gamma}\max_{t\in[0,1]}I(\gamma(t))\ \ {\rm and}\ \ \Gamma=\{\gamma\in C([0,1];W): \gamma(0)=0,\gamma(1)=u^*\}.
\end{align*}
Since $I$ is coercive in $W$ by Lemma \ref{lemma3.1}, the sequence $\{u_n\}$ is bounded in $W$. By the reflexivity of $W$, up to a subsequence, still denoted by $\{u_n\}$ such that $u_n\rightharpoonup u$ weakly in $W$. Then $\langle I^\prime(u_n),u_n-u\rangle\rightarrow0$. Thus, we obtain
\begin{align}\label{l9}
&\langle I^\prime(u_n),u_n-u\rangle\nonumber\\
=&\int_{\mathbb{R}^{2N}}\frac{ |u_n(x)-u_n(y)|^{p-2}(u_n(x)-u_n(y))(u_n(x)-u(x)-u_n(y)+u(y))}{|x-y|^{N+ps}}dxdy\nonumber\\
&+\int_{\mathbb{R}^N}V(x)|u_n(x)|^{p-2}u_n(x)(u_n(x)-u(x))dx+\int_{\mathbb{R}^N}b(x)|u_n(x)|^{q-2}u_n(x)(u_n(x)-u(x))dx\nonumber\\
  &-\lambda\int_{\mathbb{R}^N}a(x) |u_n(x)|^{r-2}u_n(x)(u_n(x)-u(x))dx
  \rightarrow0,
\end{align}
as $n\rightarrow\infty$. By Theorem \ref{th4}, up to a subsequence,
\begin{align*}
u_n\rightarrow u\ \ {\rm strongly\ in}\ L^r(\mathbb{R}^N,a)\ \ {\rm and\ \ a.e.\ in}\ \mathbb{R}^N.
\end{align*}
Thus, $\int_{\mathbb{R}^N}a(x)|u_n(x)|^{r-2}u_n(x)(u_n(x)-u(x))dx\rightarrow 0$ as $n\rightarrow\infty$.
Furthermore, \eqref{l9} and the weak convergence of $u_n$ in $W$ imply that
\begin{align}\label{l9.1}
&\int_{\mathbb{R}^{2N}}\frac{|u_n(x)-u_n(y)|^{p-2}(u_n(x)-u_n(y))(u_n(x)-u(x)-u_n(y)+u(y))}{|x-y|^{N+ps}}dxdy\nonumber\\
&-\int_{\mathbb{R}^{2N}}\frac{|u(x)-u(y)|^{p-2}(u(x)-u(y))(u_n(x)-u(x)-u_n(y)+u(y))}{|x-y|^{N+ps}}dxdy\nonumber\\
&\longrightarrow0,
\end{align}
as $n\rightarrow\infty$ and
\begin{align}\label{l9.2}
&\int_{\mathbb{R}^N}V(x)\left[|u_n(x)|^{p-2}u_n(x)-|u(x)|^{p-2}u(x)\right](u_n(x)-u(x))dx\nonumber\\
&+\int_{\mathbb{R}^N}b(x)\left[|u_n(x)|^{q-2}u_n(x)-|u(x)|^{q-2}u(x)\right](u_n(x)-u(x))dx\rightarrow0,
\end{align}
as $n\rightarrow\infty$. Note that the well-known vector inequalities:
\begin{align}\label{001}
&\left(|\xi|^{p-2}\xi-|\eta|^{p-2}\eta\right)\cdot(\xi-\eta)\geq C_{p}|\xi-\eta|^{p},\ \ p\geq2; \nonumber\\
&\left(|\xi|^{p-2}\xi-|\eta|^{p-2}\eta\right)\cdot(\xi-\eta)\geq \widetilde{C}_{p}\frac{|\xi-\eta|^2}{(|\xi|+|\eta|)^{2-p}},\ \ 1<p<2,
\end{align}
for all $\xi,\eta\in\mathbb{R}^N$, where $C_{p}, \widetilde{C}_{p}$ are positive constants depending only on $p$.
By \eqref{001} and \eqref{l9.1}, we obtain for $p\geq2$
\begin{align}\label{l10}
&\int_{\mathbb{R}^{2N}}|u_n(x)-u_n(y)-u(x)+u(y))|^p|x-y|^{-(N+ps)}dxdy\nonumber\\
\leq& C_{p}^{-1}\int_{\mathbb{R}^{2N}}\left[|u_n(x)-u_n(y)|^{p-2}(u_n(x)-u_n(y))-|u(x)-u(y)|^{p-2}(u(x)-u(y))\right]\nonumber\\
&\ \ \ \ \ \ \ \ \ \ \ \ \times\left(u_n(x)-u(x)-u_n(y)+u(y)\right)|x-y|^{-(N+ps)}dxdy\nonumber\\
  &\longrightarrow0,
\end{align}
as $n\rightarrow\infty$ and for $1<p<2$
\begin{align}\label{l11}
&\int_{\mathbb{R}^{2N}}|u_n(x)-u_n(y)-u(x)+u(y))|^p|x-y|^{-(N+ps)}dxdy\nonumber\\
\leq&\widetilde{C}_{p}^{-p/2}\bigg\{\int_{\mathbb{R}^{2N}}\left[|u_n(x)-u_n(y)|^{p-2}(u_n(x)-u_n(y))-|u(x)-u(y)|^{p-2}(u(x)-u(y))\right]\nonumber\\
&\ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \times\left(u_n(x)-u(x)-u_n(y)+u(y)\right)|x-y|^{-(N+ps)}dxdy\bigg\}^{p/2}\nonumber\\
&\ \ \ \ \ \ \ \ \ \ \times\left\{\int_{\mathbb{R}^{2N}}(|u_n(x)-u_n(y)|+|u(x)-u(y)|)^p|x-y|^{-(N+ps)}dxdy\right\}^{(2-p)/2}\nonumber\\
\leq& C\bigg\{\int_{\mathbb{R}^{2N}}\left[|u_n(x)-u_n(y)|^{p-2}(u_n(x)-u_n(y))-|u(x)-u(y)|^{p-2}(u(x)-u(y))\right]\nonumber\\
&\ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \times\left(u_n(x)-u(x)-u_n(y)+u(y)\right)|x-y|^{-(N+ps)}dxdy\bigg\}^{p/2}\nonumber\\
  &\longrightarrow0,
\end{align}
as $n\rightarrow\infty$, where $C>0$ is a constant.
Similarly, using \eqref{l9.2}, we can deduce
\begin{align}\label{l11.1}
\lim_{n\rightarrow\infty}\int_{\mathbb{R}^N}V(x)|u_n(x)-u(x)|^pdx=\lim_{n\rightarrow\infty}\int_{\mathbb{R}^N}b(x)|u_n(x)-u(x)|^qdx=0.
\end{align}
Combining \eqref{l10}, \eqref{l11} with \eqref{l11.1}, we get that $u_n\rightarrow u$ strongly in $W$ as $n\rightarrow\infty$.

A similar discussion as in Lemma \ref{lemma3.2}, we have
\begin{align}\label{l11.2}
&\lim_{n\rightarrow\infty}\int_{\mathbb{R}^{2N}}\left|\frac{ |u_n(x)-u_n(y)|^{p-2}(u_n(x)-u_n(y))-|u(x)-u(y)|^{p-2}(u_n(x)-u_n(y))}{|x-y|^{(N+ps)/p^\prime}}\right|^{p^\prime}dxdy\nonumber\\
&=0,
\end{align}
\begin{align}\label{l11.3}
 \lim_{n\rightarrow\infty}\int_{\mathbb{R}^N}V(x)\left||u_n(x)|^{p-2}u_n(x)-|u(x)|^{p-2}u(x)\right|^{p^\prime}dx=0,
\end{align}
and
\begin{align}\label{l11.4}
 \lim_{n\rightarrow\infty}\int_{\mathbb{R}^N}b(x)\left||u_n(x)|^{q-2}u_n(x)-|u(x)|^{q-2}u(x)\right|^{q^\prime}dx=0.
\end{align}
Clearly, for any $\varphi\in W$
\begin{align*}
\langle I^\prime(u_n),\varphi\rangle
=&\int_{\mathbb{R}^{2N}}\frac{ |u_n(x)-u_n(y)|^{p-2}(u_n(x)-u_n(y))(\varphi(x)-\varphi(y))}{|x-y|^{N+ps}}dxdy\nonumber\\
&+\int_{\mathbb{R}^N}V(x)|u_n(x)|^{p-2}u_n(x)\varphi(x)dx+\int_{\mathbb{R}^N}b(x)|u_n(x)|^{q-2}u_n(x)\varphi(x)dx\nonumber\\
  &-\lambda\int_{\mathbb{R}^N}a(x)|u_n(x)|^{r-2}u_n(x)\varphi(x)dx\nonumber\\
  &\longrightarrow0,
\end{align*}
as $n\rightarrow\infty$. By \eqref{l11.2}--\eqref{l11.4}, we get
\begin{align*}
\langle I^\prime(u),\varphi\rangle=0
\end{align*}
for all $\varphi\in W$, that is, $u$ is a critical point of $I$ in $W$. Moreover, by the continuity of $I$ and strong convergence of $u_n\rightarrow u$ in $W$, we obtain
\begin{align*}
I(u)=\lim_{n\rightarrow\infty}I(u_n)=c>0.
\end{align*}
This completes the proof.
\end{proof}


\begin{proof}[\bf Proof of Theorem \ref{Th1}]
(i) We only need to verify that if $u\in W\setminus \{0\}$ is a weak solution for problem \eqref{k1}, then there exists $\lambda_0>0$ such that
$\lambda\geq \lambda_0$. For this, taking $\varphi=u$, we obtain
\begin{align}\label{3.25}
&\int_{\mathbb{R}^{2N}}\frac{|u(x)-u(y)|^{p}}{|x-y|^{N+ps}}dxdy
+\int_{\mathbb{R}^N}V(x)|u(x)|^pdx+\frac{1}{q}\int_{\mathbb{R}^N}b(x)|u(x)|^qdx\nonumber\\
&=\lambda\int_{\mathbb{R}^N}a(x)|u(x)|^rdx.
\end{align}
Then $0<\min\{1,V_0\}\|u\|^p_{W^{s,p}(\mathbb{R}^N)}\leq \lambda\|u\|^r_{L^r(\mathbb{R}^N,a)}$ by \eqref{3.25}. So that $\lambda>0$.
By applying the inequality \eqref{elementary}, we have
\begin{align}\label{3.26}
\lambda a(x)|u|^{r-p}-b(x)|u|^{q-p}\leq \lambda^{(q-p)/(q-r)}\left[\frac{a(x)^{q-p}}{b(x)^{r-p}}\right]^{1/(q-r)}\ \ {\rm for\ all}\ x\in\mathbb{R}^N.
\end{align}
It follows from \eqref{3.25}, \eqref{3.26} and \eqref{embedding} that
\begin{align*}
&\min\{1,V_0\}\|u\|_{W^{s,p}(\mathbb{R}^N)}^p\\
&\leq \lambda^{(q-p)/(q-r)}\int_{\mathbb{R}^N}\left[\frac{a(x)^{q-p}}{b(x)^{r-p}}\right]^{1/(q-r)}|u|^pdx\\
&\leq C^p_*\lambda^{(q-p)/(q-r)}\left\|\left[\frac{a(x)^{q-p}}{b(x)^{r-p}}\right]^{1/(q-r)}\right\|_{L^{\frac{N}{ps}}(\mathbb{R}^N)}\|u\|_{W^{s,p}(\mathbb{R}^N)}^p.
\end{align*}
Thus,
\begin{align*}
\left(\min\{1,V_0\}-C^p_*\lambda^{(q-p)/(q-r)}\left\|\left[\frac{a(x)^{q-p}}{b(x)^{r-p}}\right]^{1/(q-r)}\right\|_{L^{\frac{N}{ps}}(\mathbb{R}^N)}\right)\|u\|_{W^{s,p}(\mathbb{R}^N)}^p\leq0,
\end{align*}
which implies that
\begin{align*}
\lambda\geq \left(\frac{\min\{1,V_0\}}{C_*^p\left\|\left[\frac{a(x)^{q-p}}{b(x)^{r-p}}\right]^{1/(q-r)}\right\|_{L^{\frac{N}{ps}}(\mathbb{R}^N)}}\right)^{(q-r)/(q-p)}:=\lambda_0.
\end{align*}
Therefore, if $u$ is a nontrivial weak solution of problem \eqref{k1}, then $\lambda\geq \lambda_0$.


(ii) Theorem \ref{th3.1} and Theorem \ref{th3.3} assure that for all $\lambda>\lambda^*$ problem \eqref{k1} admits two nontrivial weak solutions in $W$ in which one has negative energy and another has positive energy.
\end{proof}

\begin{corollary}
Suppose that all the assumptions of Theorem \ref{Th1} are satisfied. Then problem \eqref{k1} at least exist two nontrivial nonnegative weak solutions in $W$ in which one has negative energy and another has positive energy.
\end{corollary}
\begin{proof}
First, for $v\in W$, we have $v^+\in W$, where $v^+=\max\{v,0\}=(|v|+v)/2$. Indeed, we have
\begin{align*}
&\int_{\mathbb{R}^{2N}}\frac{|v^+(x)-v^+(y)|^p}{|x-y|^{N+ps}}dxdy\\
=& \int_{\mathbb{R}^{2N}}\left|\frac{|v(x)|-|v(y)|+v(x)-v(y)}{2}\right|^p|x-y|^{-N-ps}dxdy\\
\leq& \int_{\mathbb{R}^{2N}}\frac{|v(x)-v(y)|^p}{|x-y|^{N+ps}}dxdy.
\end{align*}
So, $v^+\in W$. Similarly, $v^-=\max\{-v,0\}$ is also in $W$.
Now, we define
\begin{align*}
I^+(u)=J(u)-H(u^+),
\end{align*}
where $H(u^+)=(\lambda/r)\int_{\mathbb{R}^N}a(x)|u^+(x)|^{r}dx$.
Then, $I^+$ is well defined on $W$ and of class $C^1$ and
\begin{align}\label{l15}
&\langle (I^+(u))^\prime,\varphi\rangle\nonumber\\
=&\int_{\mathbb{R}^{2N}}\frac{ |u(x)-u(y)|^{p-2}(u(x)-u(y))(\varphi(x)-\varphi(y))}{|x-y|^{N+ps}}dxdy\nonumber\\
&+\int_{\mathbb{R}^N}V(x)|u(x)|^{p-2}u(x)\varphi(x)dx+\int_{\mathbb{R}^N}b(x)|u(x)|^{q-2}u(x)\varphi(x)dx\nonumber\\
&-\lambda\int_{\mathbb{R}^N}a(x) |u^+(x)|^{r-2}u^+(x)\varphi(x)dx.
\end{align}
Moreover, $I^+$ is coercive, weakly lower semi-continuous in $W$ and $I^+(0)=0$. Notice that all critical points of $I^+$ are nonnegative. Indeed, if $u$ is a critical point of $I^+$, then by \eqref{l15} we have
\begin{align}\label{l16}
&\langle (I^+(u))^\prime,u^-\rangle\nonumber\\
=&\int_{\mathbb{R}^{2N}}\frac{ |u(x)-u(y)|^{p-2}(u(x)-u(y))(u^-(x)-u^-(y))}{|x-y|^{N+ps}}dxdy\nonumber\\
&+\int_{\mathbb{R}^N}V(x)|u(x)|^{p-2}u(x)u^+(x)dx+\int_{\mathbb{R}^N}b(x)|u(x)|^{q-2}u(x)u^-(x)dx\nonumber\\
&-\lambda\int_{\mathbb{R}^N} a(x)|u^+(x)|^{r-2}u^+(x)u^-(x)dx=0.
\end{align}
It follows from \eqref{l16} that $u^-=0$ a.e. in $\mathbb{R}^N$. Thus, $u\geq0$ a.e. in $\mathbb{R}^N$.

Similar to Theorem \ref{th3.1}, there exists $0\leq u^{**}\in W$ such that
$$I^+(u^{**})=\inf_{u\in W}I^+(u).$$
It is easy to see that $u^{**}\neq0$ and $I^+(u^{**})<0$. Hence, we get a nontrivial nonnegative weak solution $u^{**}$ of \eqref{k1} with negative energy.


Next, we prove that \eqref{k1} admits a nontrivial nonnegative weak solution in $W$ with positive energy. Clearly, $I^+$ satisfies Lemma \ref{lemma3.5}. We know that $I^+$
satisfies the geometrical structure of Theorem \ref{th3.2} by taking $e=u^{**}$. Similar to Theorem \ref{th3.3}, we get that $I^+$ has a critical point $0\leq u_+$ in $W$ satisfying $I^+(u_+)>0$. Therefore, this corollary is proved.
\end{proof}



\noindent\textbf{Acknowledgments.} The authors would like to express their sincere thanks to Professor Patrizia Pucci
for valuable suggestions on the topic of this paper.
M. Xiang was supported by Fundamental Research Funds for the Central Universities (No. 3122015L014).
B. Zhang was supported by Natural Science Foundation of Heilongjiang Province of China (No. A201306) and Research Foundation of Heilongjiang Educational Committee (No. 12541667) and Doctoral Research Foundation of Heilongjiang Institute of Technology (No. 2013BJ15). V.~R\u adulescu acknowledges the support through Grant CNCS PCCA-23/2014.





\end{document}